Author's response to reviews

Title: Methodology of the Prospective Economic Evaluation of the Prophylaxis for ThromboEmbolism in Critical Care Trial (E-PROTECT)

Authors:

Robert A Fowler (rob.fowler@sunnybrook.ca)
Nicole Mittmann (nicole.mittmann@sri.utoronto.ca)
William Geerts (William.geerts@sunnybrook.ca)
Diane Heels-Ansdell (ansdell@mcmaster.ca)
Michael K Gould (Michael.K.Gould@kp.org)
Gordon Guyatt (guyatt@mcmaster.ca)
Murray Krahn (murray.krahn@theta.utoronto.ca)
Simon Finfer (sfinfer@georgeinstitute.org.au)
Ruxandra Pinto (ruxandra.pinto@sunnybrook.ca)
Brian Chan (chunfai.chan@mail.utoronto.ca)
Orges Ormanidhi (orgesormanidhi@gmail.com)
Yaseen Arabi (yaseenarabi@yahoo.com)
Ismael Qushmaq (iqushmaq@kfshrc.edu.sa)
Marcelo G Rocha (m.garcia.rocha@gmail.com)
Peter Dodek (peter.dodek@ubc.ca)
Lauralyn McIntyre (lmcintyre@ottawahospital.on.ca)
Richard Hall (R.I.Hall@dal.ca)
Niall D Ferguson (n.ferguson@utoronto.ca)
Sangeeta Mehta (Geeta.mehta@utoronto.ca)
John C Marshall (marshallj@smh.ca)
Christopher J Doig (cdoig@ucalgary.ca)
John Muscedere (muscedej@KGH.KARI.NET)
Michael J Jacka (mjacka@ualberta.ca)
James R Klinger (James_klinger@brown.edu)
Nicholas Vlahakis (n08y@ymail.com)
Neil Orford (neilo@barwonhealth.org.au)
Ian Seppelt (ian.seppelt@sydney.edu.au)
Yoanna K Skrobik (skrobik@sympatico.ca)
Sachin Sud (Sachinsud@aol.com)
Jack F Cade (Jack.Cade@mh.org.au)
Jamie Cooper (jamie.cooper@monash.edu)
Deborah J Cook (debcook@mcmaster.ca)

Version: 5 Date: 16 November 2014

Author's response to reviews:

Dear Dr Fowler,

Thank you for uploading your revised manuscript. Your paper was re-reviewed,
and the handling editor has completed his assessment of the report and your revisions. Before we proceed with your submission, please address the following formatting requests:

1. Please ensure the title fits journal style for study protocol articles as closely as possible. The title should follow the format “____________: study protocol for a randomized controlled trial.” Your title should at least include ‘study protocol’.
   > Done.

2. Please remove the contributors who do not qualify for authorship from your title page, and include them (with their permission) in the Acknowledgements section if they are not already present here.
   > Done

3. Please include the date of registration with the trial registration number at the end of the Abstract.
   > Done. September 10, 2005

4. Please include the names of all ethical bodies that approved your study in the various centres involved, along with their reference numbers, in the Methods section. If you do not wish to list them all in the Methods section, please include the list as an additional file and refer to this in the methods section.
   > This is included.

5. Please move your funding statement to the Acknowledgements section.
   > Done.

6. Please remove the attached tracked changes version of your manuscript.
   > Done

We would be grateful if you would make the changes detailed above and upload a revised version of your manuscript to the online system. Alternatively, if you would prefer to send the revised manuscript to this address, we will upload it on your behalf.

> I have upload here - thank you.

Please don’t hesitate to contact us should you have any questions.

Best wishes,

Jessie McGavin
Editorial Assistant